Author's response to reviews

Title: The effect of total arterial grafting on medium-term outcomes following coronary artery bypass grafting

Authors:

Jean-Francois Legare (jean.legare@cdha.nshealth.ca)
Ansar Hassan (ahassan@dal.ca)
Karen J Buth (kbuth@dal.ca)
John A Sullivan (sullivani@cdha.nshealth.ca)

Version: 4 Date: 12 October 2007

Author's response to reviews: see over
Dear Editor,

This letter is being written on behalf of the participating authors with regards to the submission to the *Journal of cardiothoracic surgery* “The effect of total arterial grafting on medium-term outcomes following coronary artery bypass surgery.”

Formatting changes were incorporated as suggested:
1) Legare et al. TAG was removed
2) Author emails were added
3) Keywords were removed
4) No Section headings problem were noted
5) Abbreviations was corrected
6) Author’s contributions was corrected
7) Acknowledgment section was added
8) The manuscript was saved in rtf format avoid problem with EndNote program
9) There are no figure 4 such that the legend was removed

The authors would like to thank you for considering the submission of this paper and look forward to hearing from you in the near future.

Sincerely,

Jean-Francois Légaré
Surgical Director of Cardiac Transplantation
Director of Research Division of Cardiac Surgery